# Personal Health Data: A Systematic Mapping Study

Maria Karampela<sup>1</sup>, Sofia Ouhbi<sup>2</sup>, Minna Isomursu<sup>1</sup>

# Abstract

*Background:* Personal Health Data (PHD) research has been intensified over the last years, attracting the attention of scientists from different fields, such as Software Engineers, Computer Scientists and Medical Professionals. The increasing interest of researchers can be attributed to the exponential growth of the available PHD due to the widespread adoption of ubiquitous technology in everyday life, as well as to the potential of the ongoing digital transformation in healthcare. This increasing interest requires that academia has an overview of the published scientific literature to plan future endeavors.

*Objective:* The main objective of this study is to identify and address research gaps in literature regarding PHD.

*Method:* This paper conducts a systematic mapping study to summarize the existing PHD approaches in literature and to organize the selected studies according to six classification criteria: publication source, publication year, research types, empirical types, contribution types and research topic.

Results: In total 79 papers have been included after fulfilling the inclusion

*Email addresses:* makar@itu.dk (Maria Karampela), sofia.ouhbi@uir.ac.ma (Sofia Ouhbi), miis@itu.dk (Minna Isomursu)

<sup>&</sup>lt;sup>1</sup>IT University of Copenhagen, Copenhagen, Rued Langgaards Vej 7 DK-2300 Copenhagen S, Denmark. Tel.: (+45) 72 18 50 00.

 $<sup>^2\</sup>mathrm{TICLab},$  FIL, International University of Rabat, Technopolis Rabat-Shore Rocade Rabat-Sal, Rabat, Morocco.

criteria and have been classified accordingly. There is an increasing amount of attention that has been paid to PHD since 2014. The majority of papers is published in journals. The two main research types found were solution proposals and evaluation research. The majority of the selected papers were empirically evaluated. The main contribution types were methods and frameworks. Data privacy is the most frequently addressed topic in PHD literature, followed by data sharing.

*Conclusions:* The findings of this systematic mapping study have implications for both researchers who are planning new studies in PHD and for practitioners who are working in connected health and would like to have an overview on the existent studies on PHD research area.

*Keywords:* Personal health data, systematic mapping study, digital data, data privacy and security, data sharing

# 1. Introduction

A demographic change has been projected worldwide, the populations are rapidly aging. According to World Health Organization (WHO), "Between 2015 and 2050, the proportion of the world's population over 60 years will nearly double from 12%, to 22%. By 2020, the number of people aged 60 years and older will outnumber children younger than 5 years" [1]. This shift in the distribution of the global population towards aging poses significant challenges to countries as they have to ensure that social and healthcare systems are aligned with the needs of older populations. Even so, as people grow older tend to develop more complex health conditions increasing the likelihood of having one or multiple chronic diseases. Traditional healthcare services are often designed to manage health conditions in disconnected and fragmented ways so that in many cases hinder the communication between patients and healthcare actors. To support this change future healthcare systems need to undergo a transformation so that can ensure effortless communication and coordination between stakeholders. Advances in digital health could shift the tide of healthcare services towards long-term and sustainable solutions.

Technological innovations in digital health have been seen to hold the potential for reformation of healthcare services. The development of information and communication technologies (ICT) and their adoption in healthcare industry is changing the provision of services, not only improving the amount and quality of them, but also facilitating effective communication between patients and medical professionals [2]. It has been foreseen that ICT will facilitate the shifting center of care from physicians to patients and to self-management of diseases. Current digital solutions such as home nursing [3], health monitoring systems [4], mobile applications [5], electronic medical records (EMR) [6], electronic health records (EHR) [7], personal health records (PHRs) [8] and mobile PHRs (mPHRs) [9], can support this argument.

In the same vein, EHRs' format standardization and expansion of access to patients is a milestone towards the promotion of self-management of diseases [8, 10]. By granting patients with access and ownership to EHR, a primary form of PHRs emerged by EHR [11]. The context of EHR is restricted to medical information and it is useful for diagnosis of diseases [12, 13], so that patients have limited contribution and control over them

[11]. In contrast to EHRs, PHRs hold health and wellness data that are exclusively controlled by patients across their life span. Besides that, PHRs can support patients to make health decisions and to self-manage their health [8, 14]. The first discussions about PHRs were held in late 1970's [15]. The original idea emerged from the need to personalize technologies and to make health records accessible to the general population. PHRs can facilitate patients to self-monitor their own health and enable them to store information about behaviors, medical treatments or allergies [16]. PHRs can stand-alone on personal computers [17] or can be a part of a system administered by healthcare organizations [8]. Besides PHRs and EHRs, mPHRs allow users to access and coordinate their life long health information through their smartphones and enable them to distribute their personal health data (PHD) when necessary [9, 18]. The system combining EHR and mPHR is considered a powerful tool that can facilitate seamless communication between patients and medical professionals leading to faster and informed decisions, especially in cases of patients with chronic conditions [19]. Unified systems combining PHRs and EHRs could maximize patients' engagement and increase their participation to shared decision-making processes [20].

Towards the same direction, the proliferation of network sensors either worn or placed in the living spaces and connected to the Internet of Things (IoT) could bring patients one step closer to personalized healthcare [21, 22]. Wearable sensors, for example smartwatches or accelerometers and gyroscopes built-in smartphones, allow real-time physiological monitoring, providing thus PHD about daily habits in real-world contexts [23]. PHD has the potential to revolutionize the landscape of healthcare services providing information that could lead to customization of medical treatments and even to diseases prediction leveraging big data classification techniques [24]. Big data analytics technology holds the potential to transform health information by harnessing complex and large amount of datasets to infer real-time knowledge and new insights from it [25, 26, 27].

PHD is a research topic that has attracted the interest of researchers and practitioners from various disciplines, such as Computer Scientists, Engineers and Medical Professionals [28, 29, 30]. However, to the best of our knowledge, no previous study has conducted a systematic mapping study on PHD. The aim of this paper is to identify and map the existence research in PHD research area, addressing possible research gaps and suggesting future research directions. In this paper, a systematic mapping study has been performed, to answer six mapping questions (MQs), resulting to the selection and classification of 79 papers. The classification criteria are the following: publication source, publication year, research types, empirical types, contribution types and research topic.

The remainder of paper is organized as follows. Section 2 provides a summary of related work concerning PHD. Section 3 outlines the research methodology of this paper. Section 4 reports the classification results, while Section 5 discusses the principle findings, presents implications for researchers and practitioners and addresses the limitations of this study. Finally, Section 6 presents the conclusions and proposes work that will be carried out in the future.

#### 2. Background and significance

Various aspects of PHD have been studied and discussed in literature. For instance, the value of visualization modalities [28], mobile-based solutions for integration of PHD to healthcare systems [31] and strategies for development of infrastructures for management of PHD [32]. Many researchers have carried out studies focusing on security and privacy issues of PHD proposing solutions for minimizing data breaches [33], or legal frameworks, policies and ethical considerations for protection of PHD [34, 35, 36]. PHD management leveraging meta-data and cloud-computing for seamless access and sharing of data among medical professionals [37], as well as access control and security of PHD in cloud-computing have also been the subject of discussions [38]. Moreover, the potential of data mining and big data techniques in healthcare have been reviewed [39, 40]. Studies have also addressed challenges concerning sensor interoperability in terms of infrastructure capabilities [41] and data fusion conflicts for the integration of sensor data to EHR [42]. Previous studies concluded that patients with long-term conditions, such as diabetes and congestive heart failure, or medical treatments as vitro fertilization, which creates new needs for information or communication, are the users that will probably visit their health records more frequently [43, 44, 45].

Over the last years researchers have been given attention to PHD, for this reason we believe that is important to present an overview of this research area. Systematic literature reviews (SLR) and systematic mapping studies have been used to give an overview of the state-of the art of different disciplines [46]. The benefits and drawbacks of conducting literature reviews have been addressed by various studies [47, 48]. In this paper, a system-

atic mapping study has been used to provide an overview of the available papers. Our aim is to classify the selected studies, based on the established classification scheme. To the best of our knowledge, no previous studies have reported and mapped PHD using this classification scheme. Nevertheless, Acher et al. [17] and Roehrs et al. [49] have conducted literature reviews on PHRs. They have presented previous literature using different classification schemes and sets of research questions. Besides that, Gagnon et al. [50] have presented a contribution about electronic PHR (ePHR) reporting federal and local laws and policies that influenced ePHR implementation in Canada. Moreover, a literature review focusing on consumer perspectives of PHR has been conducted [51].

#### 3. Methodology

The systematic mapping study principal goal is to provide an overview of a research area, and identify the quantity and type of research and results available within it. A mapping process consists of three activities: the search for relevant publications, the definition of a classification scheme and the mapping of publications [52]. This method focuses on classification, conducting a thematic analysis and identifying publication fora. Fig. 1 shows the mapping process, which covers the search for relevant publications, the definition of a classification scheme, and the mapping of publications.

#### 3.1. Mapping questions

This study aims to gain insight into the existing literature about PHD. The systematic mapping study therefore addresses six MQs. The MQs with the rationale motivating the importance of these questions are presented in



#### Figure 1: Systematic mapping process

ID	Mapping question	Rationale
MQ1	Which publication channels are the main tar-	To identify where PHD research can be found
	gets for PHD research?	as well as the good targets for publication of
		future studies.
MQ2	How has the frequency of studies related to	To identify the publication trends over time of
	PHD changed over time?	PHD literature.
MQ3	What are the research types of PHD studies?	To explore the different types of research re-
		ported in the literature concerning PHD.
MQ4	Are PHD studies empirically validated?	To discover whether research on PHD has been
		validated through empirical studies.
MQ5	What are the approaches that were reported	To discover the existing PHD approaches re-
	in PHD research?	ported in the existing PHD literature.
MQ6	What are the main topics in PHD literature?	To identify the research areas related to PHD.

Table 1:	Mapping	questions
----------	---------	-----------

Table 1. The search strategy and paper selection criteria were defined on the basis of them.

# 3.2. Search strategy

The papers were identified by consulting the following sources: IEEE Digital Library, ACM Digital Library, Science Direct and SpringerLink. Google scholar was also used to seek literature in the field. The search was performed in January 2018. The search string used to perform the automatic research in the digital libraries selected was formulated as follows: "Personal" **AND**  "health" **AND** "data". This search string was applied in the title, abstract and keywords of the investigated papers to reduce the search results. The search string was limited to three keywords as the authors' search strategy was to include a broad selection of relevant literature in the first iteration, relying afterwards on the screening of papers to include or exclude them in the study [46].

#### 3.3. Paper selection criteria

Each paper was retrieved by the first author and the information about it was filed in an excel file. All results from the search engines were retrieved and analysed. The author considered each paper's title, abstract and keywords, and commented on whether the paper should be included or excluded in the excel file. The second author revised the final selection. The first step after the application of the search string was to eliminate duplicate titles, and titles clearly not related to the review. The inclusion criteria (IC) were limited to:

IC The studies that address personal health data.

The studies that met at least one of the following exclusion criteria (EC) were excluded:

EC1 Papers that focus on personal pets.

EC2 Papers whose subject is non-digital health data.

EC3 Papers that focus on systems for data collection.

EC4 Papers that focus on information systems interoperability.

EC5 papers that focus on systems for data management.

EC6 Papers whose subjects is law.

The authors excluded papers that were related to pets (EC1), as well as studies that employed non-digital methods to collect PHD (EC2). Besides that, were also excluded studies concerning the assessment of medical information systems, namely studies that were focused only on the technical aspects of healthcare systems (EC3) and information systems interoperability (EC4). Likewise, out of the scope of this systematic mapping study was the investigation of practices in health management (EC5) and legal frameworks (EC6) as they are contextual to business-oriented studies.

In total, 246 papers were identified after the removal of duplicates. When the same paper appeared in more than one source, it was considered only once according to our search order. Thereafter, 167 studies were excluded based on the inclusion and exclusion criteria leaving for the final result 79 selected studies.

#### 3.4. Data extraction strategy

The selected studies were exploited to collect the data that would provide the set of possible answers to the MQs. The publication source and channel of the papers selected respond to **MQ1**, while the publication year responds to **MQ2**.

MQ3. A research type can be classified in the following categories [53]:

• Evaluation research: Existing PHD approaches are implemented in practice and an evaluation of them is conducted.

- Solution proposal: A PHD solution is proposed. This solution may be a new PHD approach or a significant extension of an existing approach. The potential benefits and the applicability of the solution could be shown with an empirical study or a good argumentation.
- Opinion paper: These papers express the personal opinion of somebody whether a certain technique is valuable or not, or how things should been done.
- Review: Analysis of PHD existing literature.
- Other, e.g. experience papers, which express the personal experience of author(s), explaining the actions and practical aspects of the topic under discussion.

MQ4. The selected studies can be classified as a [54, 55]:

- Case study: An empirical inquiry that investigates a PHD approach within its real-life context.
- Survey: A method for collecting quantitative information concerning a PHD approach, e.g. a questionnaire.
- Experiment: An empirical method applied under controlled conditions, in order to evaluate a PHD approach.
- History-based evaluation: Studies evaluating PHD approaches in previously completed software projects.
- Theory: Non-empirical research approaches or theoretical evaluation of a PHD approach.

MQ5. An approach can be classified as [52, 56]:

- Process: A series of actions, or functions leading to a PHD result and performing operations on data.
- Method: A regular and systematic means of accomplishing PHD.
- Tool-based technique: A technique based on a software tool to accomplish PHD tasks.
- Model: A system representation that allows PHD to be investigated through a hierarchical structure.
- Framework: A real or conceptual structure intended to serve as a support or guide for PHD.
- Other, e.g. guidelines, data mining technique.

MQ6. A PHR-related topic can be classified as: PHD security, PHD privacy, PHD accessibility and PHD analysis among others that can be identified from the selected studies. To identify the main research topics of the papers, the authors relied on the analysis of the title and keywords. In cases of papers with two main topics, the authors classified the papers keeping both topics.

### 3.5. Synthesis method

The synthesis method was based on:

1. Counting the number of papers per publication channel and the number of papers found in each bibliographic source per year,

- 2. Counting the primary studies that are classified in each MQ's response,
- 3. Presenting charts for the classification results which have been used in the analysis,
- 4. Presenting in the discussion a narrative summary with which to recount the principal findings.

# 4. Results

This section describes the results related to the systematic MQs presented in Table 1. Table 2 and Table 3 give an overview of the classification results.

4.1. MQ1. Which publication channels are the main targets for PHD research?

62% of the selected papers are published in scientific journals, while 28% are in conferences. Only few studies are published in workshops, symposia, books or other channels of publication. Table 4 presents publication sources which has published more than one paper included in this study. The four sources identified are all journals. The rest of papers not shown in Table 4 were published in different venues and journals.

4.2. MQ2. How has the frequency of studies related to PHD changed over time?

Fig. 2 presents the number of articles published per year from 1991 to 2017. As Fig. 2 shows, the number of publications was less than 3 papers per year in the period from 1991 to 2013, except the years 2009 and 2012, while from 2014 the number of publications rises steadily.

Table 2: Classification results. Part 1. Acronym: History-based evaluation (	HbE)	
--	------	--

Ref.	P. Channel	Year	Research Type	Empirical Type	Contribution Type	Research topic
[57]	Journal	1991	Opinion paper	No	Guidelines	Data protection
[58]	Journal	1995	Solution proposal	Survey	Framework	Data privacy & security
[59]	Journal	1996	Review	No	Guidelines	Data security
[60]	Journal	2001	Opinion paper	No	Guidelines	Data privacy
[61]	Journal	2003	Review	No	Method	Data use
[62]	Conference	2005	Solution proposal	No	Framework	Data representation
[63]	Conference	2006	Solution proposal	No	Model	Sensor data
[35]	Journal	2007	Evaluation research	Survey	Guidelines	Data privacy
[64]	Journal	2007	Opinion paper	No	Guidelines	Data security
[65]	Journal	2007	Evaluation research	Survey	Method	Data use
[66]	Journal	2008	Evaluation research	HbE	Tool	Data sharing
[41]	Conference	2008	Solution proposal	Experiment	Framework	Sensor data
[67]	Journal	2009	Solution proposal	Other	Framework	Data integration
[68]	Conference	2009	Solution proposal	No	Tool	Data visualization
[69]	Conference	2009	Solution proposal	Experiment	Tool	Data visualization
[33]	Workshop	2009	Solution proposal	Experiment	Method	Data security
[42]	Journal	2009	Solution proposal	Experiment	Method	Data processing
[70]	Journal	2009	Evaluation research	Survey	Method	Data privacy
[38]	Conference	2010	Solution proposal	No	Framework	Data privacy & security
[71]	Journal	2010	Evaluation research	Survey	Model	Data sharing
[72]	Conference	2011	Solution proposal	Experiment	Framework	Data sharing
[73]	Conference	2011	Solution proposal	Experiment	Method	Data accessibility
[74]	Journal	2012	Review	No	Method	Data privacy & security
[75]	Journal	2012	Evaluation research	Experiment	Tool	Data accessibility
[76]	Journal	2012	Evaluation research	Survey	Method	Health records
[77]	Journal	2012	Evaluation research	Survey	Model	Data sharing
[78]	Journal	2012	Solution proposal	Case study	Framework	Sensor data
[79]	Other	2012	Evaluation research	Experiment	Method	Data privacy
[80]	Journal	2012	Opinion paper	No	Guidelines	Data accessibility
[81]	Journal	2013	Solution proposal	No	Model	Data protection
[82]	Conference	2013	Solution proposal	Other	Framework	Sensor data
[28]	Conference	2014	Evaluation research	Experiment	Method	Data visualization
[83]	Symposium	2014	Solution proposal	Survey	Framework	Data sharing
[84]	Conference	2014	Solution proposal	No	Tool	Data management
[85]	Journal	2014	Opinion paper	No	Guidelines	Sensor data
[31]	Conference	2014	Solution proposal	Experiment	Tool	Data integration
[86]	Conference	2014	Evaluation research	Survey	Method	Data sharing
[87]	Conference	2014	Solution proposal	Experiment	Framework	Data portability
[88]	Conference	2014	Solution proposal	Experiment	Framework	Data design
[29]	Journal	2014	Review	No	Method	Data accessibility
[89]	Journal	2014	Solution proposal	No	Framework	Data sharing
[90]	Journal	2014	Opinion paper	No	Guidelines	Data protection & privacy
[91]	Journal	2015	Solution proposal	No	Model	Data management
[92]	Journal	2015	Evaluation research	Survey	Method	Data sharing
[93]	Journal	2015	Opinion paper	No	Guidelines	Data security
[94]	Journal	2015	Review	No	Guidelines	Health Informatics & Edu-
						cation

Ref.	P. Channel	Year	Research Type	Empirical Type	Contribution Type	Research topic
[95]	Workshop	2015	Evaluation research	Experiment	Tool	Sensor data
[96]	Journal	2015	Review	No	Method	Data protection
[97]	Other	2015	Evaluation research	Experiment	Framework	Data privacy
[34]	Conference	2015	Review	No	Method	Data protection
[98]	Workshop	2015	Solution proposal	Experiment	Framework	Data sharing
[40]	Journal	2015	Review	No	Method	Data management
[99]	Journal	2015	Solution proposal	Case study	Tool	Data management
[100]	Journal	2016	Evaluation research	Survey	Method	Data sharing
[101]	Journal	2016	Evaluation research	Survey	Method	Data sharing
[39]	Journal	2016	Solution proposal	Experiment	Method	Data indexing
[102]	Journal	2016	Solution proposal	Experiment	Tool	Data privacy & security
[103]	Conference	2016	Solution proposal	Survey	Framework	Data analysis
[104]	Workshop	2016	Review	No	Method	Data use
[32]	Conference	2016	Evaluation research	Survey	Framework	Data design
[105]	Journal	2016	Review	No	Guidelines	Data privacy & security
[106]	Journal	2016	Solution proposal	No	Method	Data privacy
[107]	Journal	2016	Evaluation research	Survey	Framework	Data management
[14]	Conference	2016	Solution proposal	No	Tool	Data management
[108]	Journal	2017	Solution proposal	No	Model	Data protection & privacy
[109]	Journal	2017	Evaluation research	Survey	Tool	Health records
[110]	Journal	2017	Review	No	Guidelines	Data privacy
[36]	Conference	2017	Opinion paper	No	Guidelines	Data privacy
[30]	Journal	2017	Evaluation research	Survey	Tool	Data accessibility
[111]	Conference	2017	Evaluation research	HbE	Model	Data privacy & security
[112]	Journal	2017	Opinion paper	No	Guidelines	Data protection
[113]	Journal	2017	Evaluation research	Experiment	Tool	Health records
[114]	Journal	2017	Evaluation research	Survey	Guidelines	Data protection & privacy
[37]	Conference	2017	Solution proposal	No	Method	Health records
[115]	Journal	2017	Solution proposal	No	Framework	Data protection
[116]	Journal	2017	Review	No	Method	Health Informatics & Edu-
						cation
[117]	Journal	2017	Evaluation research	Survey	Framework	Health records
[118]	Journal	2017	Evaluation research	Experiment	Framework	Data protection & privacy
[119]	Journal	2017	Evaluation research	Case study	Model	Data management

Table 3: Classification results. Part 2. Acronym: History-based evaluation (HbE)

Table 4: Publication sources which published more than one PHD selected paper

Publication source	References	No.
Journal of Medical Internet Research	[66, 30, 100, 71]	4
European Journal of Risk Regulation	[112, 81]	2
International Journal of Medical Informatics	[107, 35]	2
Studies in Health Technology and Informatics	[108, 99]	2







Figure 3: MQ4, MQ5 and MQ6 results

# 4.3. MQ3. What are the research types of PHD studies?

Fig. 3 shows the research type of the selected papers. Around 41% of the selected papers were solution proposal studies, 33% of the selected papers were undertaken to evaluate PHD existing approaches, 11% were reporting the authors' opinions regarding PHD and the remaining papers were classified as reviews. Fig. 3 shows also that 41% of the solution proposals were empirically validated and that the majority of the suggested solutions are methods.

# 4.4. MQ4. Are PHD studies empirically validated?

Fig. 3 shows if the selected studies were empirically validated, and presents the empirical types used in the validation of PHD approaches. 33% of the selected studies were not evaluated empirically. 24% of the selected papers were evaluated with experiments, 24% used surveys, 4% undertook case studies and 3% used history-based evaluation.

# 4.5. MQ5. What are the approaches that were reported in PHD research?

Fig. 3 presents the approaches used for PHD in the selected papers. The approaches most frequently reported are methods (29% of the selected papers) followed by frameworks (25%). Guidelines, tool-based techniques and models were also identified in the selected studies.

# 4.6. MQ6. What are the main topics in PHD literature?

Fig. 4 shows that the main research topic of the selected papers is data privacy, followed by data sharing and data security. Other research topics were also identified in PHD literature and are presented in Fig. 5.

#### 5. Discussion

This section discusses the results and main findings of this paper. Recommendations for future research are also presented.

# 5.1. Principal findings

The main goal of this systematic mapping study is to examine the current research in PHD. The principal findings of this study are the following:



Figure 4: Research topics of the selected papers



Figure 5: Research topics in selected studies about Personal Health Data

#### 5.1.1. Publication channels

Publication channels are informative about the process of development of PHD research area. The results obtained in this study show that PHD publications are mainly published in scientific journals (62%). Scientific publications in journals have been seen as an elaborate process in terms of review criteria, acceptance practices and manuscript-journal "fit", so that the contribution of scientific publications in prestigious journals is not disputable [120, 121]. Furthermore, the visibility of publications in peer-reviewed journals is considered to be higher than other publication media [122]. Publication of peer reviewed studies in high-impact journals like JMIR, BMC and JMS, are indications of the latest growth of this research area across disciplines.

In opposition to journals, conferences have been characterized as innovationladen, while are faster publication venues for dissemination of research results, promoting also knowledge exchange [123]. The discussion regarding the empirical types of studies bellow, will further strengthen this argument discussing the current types of research in this research area. The frequency of published studies over time, will be discussed extensively in the following subsection, but a primary observation is that PHD research is a relatively new scientific research area that started progressing over the last decade. An indicator of the recent development of this research area might be attributed to the lack of EHRs interoperability standards. According to WHO, 47% of the Members States worldwide have national EHRs, with the majority of them stating interoperability as one of the main challenges for the implementation of EHRs. The adoption rates of EHRs reported over 50% for upper-middle and high-income countries, while for the lower-middle and lowincome countries the rates are lower, 35% and 15% respectively [124, 125]. Nevertheless, the adoption of PHR found to dependent on growth in EHR adoption [17].

#### 5.1.2. Publication trend

The present study classified publications from 1991 to 2017. The PHD research has attracted increasing attention since 2007. The lack of publications between 1991 and 2013 is reasonable as PHRs and EHRs are relatively recent developments in digital health. A primary definition of PHD was given by Markle Foundation's Personal Health Working Group was in 2003 [126], while in 2005 in a working symposium, the College of Medical Informatics formalized a definition of PHD [8]. Later that year, the American Health Information Management Association (AHIMA) launched a Work Group to study the role of PHD in EHR [127]. Also, the AHIMA's Foundation program "Better health information for all" started in 2009.

Personal health organizers like HealthSpace became available to all patients in UK in 2003, facilitating them to store personal health notes [128]. The adoption of EHRs in a national level in US, Australia and UK was initiated in 2004 [12]. While one of the first on-line communities for patients, the social platform PatientsLikeMe, was launched in 2006 [129], well-known PHRs such as Microsoft HealthVault [130] and Google Health [131] were released in 2007 and 2008 respectively. Nevertheless, Google's decision to stop supporting Google Health application indicates that the popularity of the product was lower than anticipated. The use of EHRs is restricted to medical information exchange systems, making evident that the integration of EHRs and PHRs is still in its infancy [125]. What considered valuable since PHRs' conception, is the ability to connect personal records with EHR, as stand alone patients' data are likely to remain outdated, acting as archives for patients' input [8, 17].

Connected health emergence within the last decade along with technological advancements in healthcare aim to change health management of businesses and organizations [132]. According to Chouvarda et al., connected health "aims for the optimal access, sharing, analysis and use of health data via systematic application of healthcare information technology, in other words to "offer the correct information to the correct person at the correct time", so that health actors (citizens, patients, clinicians, policy makers) make better decisions for health and care" [133]. Connected health includes e-Health, m-Health, telemedicine and telehealth solutions and refers also to the provision of health services remotely, with applications to the global aging population and to patients with chronic conditions [134]. The emergence of connected health model is bounded to a shift in global demographics; the share of aging population projected to increase surpassing young people [1]. The pressure of the aging population to global economy [135], comes as a result of an increasing population with chronic and degenerative diseases [1].

Reflecting upon these, we could argue that the growth of scientific publications in PHD area from 2014 onwards, portrays the demand for provision of better self-care services and adoption of agile solutions to facilitate the daily life of population. Besides that, the adoption of technological advancements such as smartphones brings patients closer than ever to personalized health care [136]. According to IDC, 344.3 million smartphones were sold worldwide in the first quarter of 2017 [137], with the number of smartphone users forecasted to reach 2.1 billion in 2020 [138], while sensors connected to IoT are projected to range from 30 to 50 billion devices [139, 140]. In agreement with these trend is also the decision of Agfa Healthcare, one of the key-players in the Healthcare IT industry to ally with My Personal Health Record Express Inc. buying 27% of the company on 2016 [141].

#### 5.1.3. Research types

The results show that the main concern of researchers in the PHD domain is to propose novel approaches or to extent existing techniques to deal with PHD. Moreover, evaluation research also is a practice used to assess previous research efforts by implementing a solution in practice, in order to provide practical explanations and to indicate advantages and disadvantages [46, 53, 142]. The results indicate that research in this area is still in its early stages, so it is essential to evaluate processes and solutions in order to evolve.

Around a third of the selected studies were not evaluation research, suggesting that the proposed approaches are not validated based on users experiences. These results suggest that future research should be focusing towards research approaches that will enable user validation in order to encapsulate users' perceptions. The value of human-centered design cannot be neglected; especially in healthcare applications users involvement throughout the design process from low-fidelity prototypes to usability testing of the final products have been seen as a crucial factor for developing valuable medical systems in connected health [143]. The Agency for Healthcare Research and Quality, as well as Food and Drug Administration (FDA) have also discussed the importance of usability and human factors evaluation of medical devices and applications during the design process [144, 145].

The following two categories, namely reviews and opinion papers are more theoretical approaches, as they do not rely on empirical validation of the proposed solutions [53, 142, 46]. The aim of these research approaches is to discuss or frame theories of PHD and to provide a theoretical framework.

#### 5.1.4. Empirical studies

The findings for this MQ show that almost half of the selected studies are not validated empirically. This means, that the studies do not provide validation of results, while the support of the proposed approaches is not based on any proof of concept. More specifically, approximately 43% of the nonempirical studies suggest theoretical approaches revealing that there is a lack of user validation on the proposed solutions in real or laboratory settings. Our results also indicate that only 24% of the selected papers were evaluated with experiments. Despite this, almost all of the studies that were classified as experiments, were conducted in laboratory settings without the participation of users [28, 69, 75]. An explanation regarding this result is related to the context of the studies. The majority of them are experimental research on technical solutions such as data mining approaches [39], cloud computing [102], or data fusion and aggregation methods for pre-processing of PHD [42]. Note also that according to our classification scheme, an experiment is an empirical method applied under controlled conditions to evaluate a PHD approach. Therefore, there is no strict constraint for evaluating the proposed solution on users. Besides that, the same percentage with experiments was classified as surveys. Surveys are methods that capture the opinion of users indirectly using instruments such as questionnaires. The low percentage of case studies also verifies the lack of user validation on the proposed solutions, as we only identify three studies: two solution proposals about sensor data and data management [78, 99], and one evaluation research which also concerns management of data [119]. The existence of these three case studies could be a good approximation of the implementation status of PHD in healthcare industry, as a case study validation requires real-world context [54, 55]. History-based evaluations also indicate that PHD research area is in an early development stage, as they point out the lack of investigation of the cases in real-world settings. Future research efforts should be focusing towards empirical validation of the proposed solutions.

#### 5.1.5. Approaches

The contribution types of the selected papers can be seen in Fig. 3. The approaches most frequently reported are methods 29 %, followed by frameworks 25%. Methods are used in research to report the steps for knowledge acquisition, while frameworks offer guidance for the development of solutions [52, 56]. The results of this mapping study show that the majority of the authors propose methods, however almost half of the cases are reviews and thus are not empirically validated [29, 34, 40, 61, 74, 96, 104, 116]. The approach of authors reveals that they only assess previous knowledge without proposing real implementation of PHD solutions. Only one study that discusses sensor technology, proposes a framework validated on data extracted from wearable sensors of a football team [78]. The contribution types of the selected papers point out a lack of user validation on the suggested solutions. The lack of user validation can be attributed to limited implementation of PHD solutions in medical industry. To our knowledge medical information systems that allow integration of EHR and PHR are modest or restricted to pilot implementations [146]. Some of the PHD approaches reported in the selected papers are listed below.

Li et al. discuss access control of PHR by users in could-computing proposing a framework for secure access control and distribution of data in systems with multiple security domains [38]. Privacy of PHD in Big data technologies in healthcare is the subject of Lee et al. study [106]. The complexity of de-identification of personal information hinders the implementation of such technologies in medical sector. The root of problem is also spotted on the lack of common agreement among parties that distribute and handle data. Pickard's study presents a global online survey about consumers' willingness to share PHD, for the creation of datasets that can predict wellness and diseases trends [86]. The survey is focusing on the influence of the financial rewards towards information sharing within the context of Markets of Data (MoDAT). Participants' willingness to share PHD has also been discussed from a different perspective by Chen et al. [101]. They concluded that participants' willingness to share personal data depends on the potential data usage. The study points out that health research and the development of targeted interventions increased participants willingness to share their data, creating thus opportunities for both researchers and industry. PHD management is a research topic that has attracted the interest of researchers and practitioners from various disciplines, such as Engineers, Computer Scientists and Medical Professionals [14, 37, 113]. Mobile-based solutions for integration of PHD to healthcare systems [31], as well as a multi-platform synchronization framework useful for speeding up the implementation of personal health systems on mobile devices [72].

# 5.1.6. Research topics

PHD is an interdisciplinary research area with contributions to various disciplines [33, 84, 115, 116]. The main research topics of the selected papers is data privacy (20%), followed by data sharing (12%) and data security (11%). What the present results indicate is that although the technology supporting PHRs is available [147], among the main concerns are privacy and security of PHD. The discussions of the selected studies are around these topics, focusing either on technical features of PHD systems like cryptography [33], cloud-computing and big data [89, 105], and security in mHealth [98] or on the legal and ethical aspects of it [57, 115]. These findings suggest that the research community approaches holistically privacy and security of PHD, identifying that both technical and human factors are essential for the adoption of sustainable solutions. Although this study is focusing on the technological aspects of PHD, the impact of the human factor on privacy and security solutions in healthcare cannot be neglected. What this study confirms is that technology alone is not enough to handle these challenges, suggesting the need for introduction of proactive legal and ethical frameworks. Technology acceptance and adoption is a topic that has been discussed on previous studies in other research areas [148], but a general principle that applies also on PHD area is that the widespread usage of PHD requires the acceptance and adoption of both medical professionals and patients.

Many of the selected studies point out that investigations are also focusing on technical and users' aspects concerning a future centralization approach of PHD (Data sharing 12%) [77, 83]. Data sharing is connected to privacy and security, as without a secure framework it would be impossible to leverage patients' PHD. Research types namely solution proposals and evaluation research, are categories with high number of publications concerning privacy and security. This indicates the need for proposition of novel solutions or significant extensions of existing approaches [149].

In contrast to data sharing, accessibility and data use were classified as less important influences on PHD area, probably because of limited real-life implementations in healthcare industry. However, access control of PHD is an issue of concern as counterweight of system preferences with security and privacy control found to be a hard task for patients [150]. Without security control over the data, patients are exposed to privacy threats [70]. The existing legal inconsistencies about personal data privacy and security will attempt to bridge the new General Data Protection Regulations (GDPR), which had been implemented in full on May 2018 [151]. The implementation of the new legislation in Europe is projected to have positive impact on data sharing in healthcare, while taking control of security and privacy issues [152]. On a worldwide scale according to WHO, 55% of the Member States have legal frameworks to protect the privacy of PHD, while 34% reported legal frameworks pertinent to sharing of PHD between health care professionals within the same country, and only 22% to sharing such data on an international level [125].

Although there has been a substantial increase in the number of policies and legislations for eHealth solutions the last decade, issues such as privacy, confidentiality and data security are among the main challenges that national healthcare systems around the world are facing as they evolve, making it clear that the issue has not only legal applications but also ethical [124, 125]. The formulation of ethical frameworks from various organizations internationally addresses the need to trigger the ethos of the scientific community. The adoption of ethical principles on decision-making pertinent to health data management can safeguard public health and lead to future sustainable solutions [153].

An interesting finding is that although many studies have addressed patients with chronic conditions and disabilities as user groups that could benefit the most from PHD utilization [154, 155, 156], only three papers [103, 113, 117] of the selected studies have empirically evaluated approaches using data from patients with chronic conditions [30].

#### 5.2. Implications

The findings of this systematic mapping study have implications for both researchers who are planning new studies in PHD and for practitioners who are working in connected health and would like to have an overview on the existent studies on PHD research area. The present study could be used as a benchmark for future research regarding PHD. Moreover, the PHD studies presented in this systematic mapping study might help researchers to identify research directions that they have not studied so far in order to improve the quality of research of this discipline.

For future research on PHD, greater presence in conferences should be considered in order to disseminate recent findings and to communicate scientific results, values and methods to researchers outside the discipline. More evaluation research should be carried out focusing on user validation in order to evaluate the proposed solutions based on real-world settings. Therefore, it is advised that future solutions should include experiments and case studies. An interesting finding is that although the existing literature has identified patients with chronic conditions as one of the target groups that could benefit from PHD adoption, in very few cases studies researchers have evaluated the proposed solutions based on data from this group of patients. Thus, in future endeavors, PHD research should be directed towards validation of research findings involving patients.

#### 5.3. Threads to validity

### 5.3.1. Construct validity

Construct threats to validity are related to the identification of primary studies [157, 158]. To limit construct threats while identifying and selecting as many as possible relevant studies, two researchers initially ran several iterations to test different strings of potential keywords. Three keywords were used in the final iteration, as the number of the returned results was broader using this research string. PHD is an interdisciplinary research field, so by using a broader search string, we have found and included publications from various research areas, limiting the possibility of excluding relevant studies. However, the list of publications might be incomplete, as alternative or additional terms might have altered the final selection of papers [159].

In this study the papers were identified consulting the following databases IEEE Digital Library, ACM Digital Library, Science Direct, SpringLink and Google scholar. The search was limited to the title, the abstract and the keywords of papers. To limit the risk of excluding relevant publications, the authors sough to use a search string limited to three keywords to include a broad selection of relevant literature in the first iteration. In the present study, the authors did not consider databases containing publications from specific regions such as Latin America and the Caribbean (Literatura Latino Americana em Cincias da Sade, LILACS), Africa (African Index Medicus, AIM) and Eastern Mediterranean Region (Index Medicus for the Eastern Mediterranean Region, IMEMR). However, the literature search has been conducted on four major global databases that are among the worlds most leading and comprehensive sources for scientific research hosting thousands of academic journals and conference papers. Also, the authors have searched for relevant literature using Google Scholar, which is a more generic bibliographic database that indexes full text or meta-data of scientific literature across disciplines. Therefore, we believe that the majority of relevant studies in the discipline has been included in this systematic mapping study. Also, including publications that are not published in high JCR impact factor journals or CORE conferences gave a proportional representation of the published research in the discipline. Due to the fact that this study is a mapping study, the exclusion and inclusion criteria are only related to whether PHD is the topic of the publication or not.

To disregard threads related to accessibility of publications due to subscription limitations of universities' libraries, both researchers agreed to mark the missing papers during the initial screening. Only two publications in total were not accessible by the authors. To overcome this barrier researchers requested the missing papers via email in ResearchGate platform. Previous research has also addressed, as a potential threat to validity, duplicated publications [160]. When duplications were detected, they were examined exhaustively to discover whether or not they were the same study. Although parts of content were common to different papers, these papers are based on novel ideas or studies. The initial selection of articles performed by the first author, while the second author revised the results. In a few cases researchers had to argue about their decision to include or exclude a publication referring back to the established inclusion and exclusion criteria.

#### 5.3.2. Internal validity

Internal validity concerns data extraction and analysis [158, 161]. The final selection and classification of articles was performed by the first two authors, while the third author revised the final version of this study and contributed addressing possible points of contention. The initial screening of papers conducted while authors were located in the same physical space, namely at IT University of Copenhagen, as the second other was conducting a research stay there during January 2018. Although authors where then located in different time zones, the initial person to person collaboration and alignment about research and data analysis methodology, contributed to the establishment of effective communication guidelines, alleviating therefore tiredness or misunderstandings due to long-distance collaboration [162]. To limit the authors' influence on the classification process, the development of the classification scheme relied on widely accepted and well-established guidelines [46]. In addition, what considered to be beneficial for the validity of the search strategy and results was the previous knowledge and skills of the second co-author on conducting systematic mapping studies [55, 56, 163, 164]. Internal validity refers also to the analysis of data. The analysis of results in this study relied only on descriptive statistics, therefore the threads to validity are minimal.

#### 5.3.3. Conclusion validity

Conclusion validity relates to imprecise conclusions that might be drawn in case of identification of unreasonable relationships. In mapping studies this thread concerns inaccurate data extraction or exclusion of relevant studies during the selection phase [161]. The control over conclusion validity has also implications to reliability of the study, so that if a systematic mapping study is performed by other researchers, they will arrive at the same conclusions [158, 159]. To handle this thread, the selection and extraction process of the data performed according to the established scheme as described in Section 3. Furthermore, the discussion of results relied exclusively on data visualization, such as bubble plots and histograms, that were generated for all of the MQs. In our viewpoint, slight differences based on publication selection bias and misclassification would have minor impact on the main conclusions drawn from the 79 publications identified in our mapping study.

# 5.3.4. External validity

This thread refers to the generalization of this mapping study [165, 166]. The results of this study are considered with regard to the PHD domain, while the validity of conclusions is applicable only on the PHD context. The external validity threads are thus not applicable. Moreover, since no time range was specified to limit the search of the published papers, the representativeness of the selected studies was not affected. Therefore, this threat is not applicable in this context. The search string and the classification scheme presented in this paper can be perceived as a baseline for future research endeavors and practitioners can search for and categorize additional papers accordingly.

#### 6. Conclusions and future work

This paper has presented a systematic mapping study that summarizes the existing research in PHD. Of 246 studies, 167 papers were identified between 1991 and 2017, 79 of which were selected and classified according to six criteria: publication source, publication year, research types, empirical types, contribution types and research topic. The findings showed that an increasing amount of attention has been paid to PHD since 2014. The two main research types found were solution proposals and evaluation research. The majority of the selected papers were empirically evaluated. Many papers proposed solutions for future research endeavors, and the main contribution types were methods and frameworks. Data privacy is the most frequently addressed topic in PHD literature, followed by data sharing. A few authors conducted research on PHD data processing or analysis. For future work, we intend to perform an SLR to identify and evaluate evidence research in PHD by taking into account the results revealed in this systematic mapping study to examine published literature in greater depth.

#### Author contributions

All authors contributed to the creation of the manuscript. MK and SO: design, conception, acquisition and interpretation of data, drafting of the manuscript, revision. MI contributed to the study by her critical revision. All authors read and approved the final manuscript. The order of authors listed in the manuscript has been approved by all of the authors.

#### Summary points

What was already known on the topic:

- There is an increasing interest about digital health hence the interest in PHD.
- No systematic mapping study has been conducted to gain insight into PHD.

What this study added to our knowledge:

- The publication trend of PHD research shows that there is an increasing interest in PHD since 2014.
- The main publication channel of PHD articles is journals.
- Solution proposals and evaluation research are the main research types of PHD literature.
- The majority of the selected papers were empirically evaluated.
- Methods and frameworks are the main approaches used in PHD literature.
- Data privacy is the most frequently addressed topic in PHD literature, followed by data sharing.

# Conflict of interest

The authors confirm that there are no known conflicts of interest associated with this publication and there has been no significant financial support for this work that could have influenced its outcome.

# **Compliance with Ethical Standards**

This article does not contain any studies with human participants or animals.

### Acknowledgements

None.

# References

[1] World Health Organization, World report on ageing and health (2015).

- [2] H. Moghaddasi, F. Asadi, A. Hosseini, Z. Ebnehoseini, E-health: a global approach with extensive semantic variation, Journal of Medical Systems 36 (2012) 3173–3176.
- [3] G.-M. Breen, N. J. Zhang, Introducing ehealth to nursing homes: theoretical analysis of improving resident care, Journal of Medical Systems 32 (2008) 187–192.
- [4] M. M. Baig, H. Gholamhosseini, Smart health monitoring systems: an overview of design and modeling, Journal of Medical Systems 37 (2013) 9898.
- [5] M. Hardinge, H. Rutter, C. Velardo, S. A. Shah, V. Williams, L. Tarassenko, A. Farmer, Using a mobile health application to support self-management in chronic obstructive pulmonary disease: a sixmonth cohort study, BMC Medical Informatics and Decision Making 15 (2015) 46.
- [6] D. A. Ludwick, J. Doucette, Adopting electronic medical records in primary care: lessons learned from health information systems implementation experience in seven countries, International Journal of Medical Informatics 78 (2009) 22–31.
- [7] P. Coorevits, M. Sundgren, G. O. Klein, A. Bahr, B. Claerhout, C. Daniel, M. Dugas, D. Dupont, A. Schmidt, P. Singleton, et al., Electronic health records: new opportunities for clinical research, Journal of Internal Medicine 274 (2013) 547–560.

- [8] P. C. Tang, J. S. Ash, D. W. Bates, J. M. Overhage, D. Z. Sands, Personal health records: definitions, benefits, and strategies for overcoming barriers to adoption, Journal of the American Medical Informatics Association 13 (2006) 121–126.
- [9] S. Ouhbi, A. Idri, J. L. Fernández-Alemán, A. Toval, Mobile personal health records for cardiovascular patients, in: Third World Conference on Complex Systems (WCCS), IEEE, 2015, pp. 1–6.
- [10] V. H. Castillo, A. I. Martínez-García, J. Pulido, A knowledge-based taxonomy of critical factors for adopting electronic health record systems by physicians: a systematic literature review, BMC Medical Informatics and Decision Making 10 (2010) 60.
- [11] C. A. Meier, M. C. Fitzgerald, J. M. Smith, eHealth: extending, enhancing, and evolving health care, Annual Review of Biomedical Engineering 15 (2013) 359–382.
- [12] T. D. Gunter, N. P. Terry, The emergence of national electronic health record architectures in the United States and Australia: models, costs, and questions, Journal of Medical Internet Research 7 (2005) e3.
- [13] S. Ouhbi, A. Idri, J. L. Fernández-Alemán, A. Toval, H. Benjelloun, Electronic health records for cardiovascular medicine, in: 36th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), IEEE, 2014, pp. 1354–1357.
- [14] J. Puustjärvi, L. Puustjärvi, Managing Fragmented Personal Data: Going beyond the Limits of Personal Health Records, in: Proceed-

ings of the International Joint Conference on Biomedical Engineering Systems and Technologies, SciTePress-Science and Technology Publications, Lda, 2016, pp. 145–150.

- [15] T. McKeown, The role of medicine: dream, mirage or nemesis?, volume 180, Taylor & Francis, 1979.
- [16] K. T. Fuji, A. A. Abbott, K. A. Galt, A. Drincic, M. Kraft, T. Kasha, Standalone personal health records in the United States: meeting patient desires, Health and Technology 2 (2012) 197–205.
- [17] N. Archer, U. Fevrier-Thomas, C. Lokker, K. A. McKibbon, S. E. Straus, Personal health records: a scoping review, Journal of the American Medical Informatics Association 18 (2011) 515–522.
- [18] H. Kharrazi, R. Chisholm, D. VanNasdale, B. Thompson, Mobile personal health records: an evaluation of features and functionality, International Journal of Medical Informatics 81 (2012) 579–593.
- [19] I.-C. Chang, S.-J. Hsiao, H.-M. Hsu, T.-H. Chen, Building mPHR to assist diabetics in self-healthcare management, in: 7th International Conference on Service Systems and Service Management (ICSSSM), IEEE, 2010, pp. 1–5.
- [20] S. Wells, R. Rozenblum, A. Park, M. Dunn, D. Bates, Personal Health records for patients with chronic disease, Applied Clinical Informatics 5 (2014) 416–429.
- [21] J. Gubbi, R. Buyya, S. Marusic, M. Palaniswami, Internet of Things

(IoT): A vision, architectural elements, and future directions, Future generation computer systems 29 (2013) 1645–1660.

- [22] S. Li, L. Da Xu, S. Zhao, The internet of things: a survey, Information Systems Frontiers 17 (2015) 243–259.
- [23] S. R. Islam, D. Kwak, M. H. Kabir, M. Hossain, K.-S. Kwak, The internet of things for health care: a comprehensive survey, IEEE Access 3 (2015) 678–708.
- [24] M. Hassanalieragh, A. Page, T. Soyata, G. Sharma, M. Aktas, G. Mateos, B. Kantarci, S. Andreescu, Health monitoring and management using Internet-of-Things (IoT) sensing with cloud-based processing: Opportunities and challenges, in: IEEE International Conference on Services Computing (SCC), IEEE, 2015, pp. 285–292.
- [25] W. Raghupathi, V. Raghupathi, Big data analytics in healthcare: promise and potential, Health information Science and Systems 2 (2014) 3.
- [26] L. M. Fernandes, M. O'Connor, V. Weaver, Big data, bigger outcomes, Journal of AHIMA 83 (2012) 38–43.
- [27] S. Kaisler, F. Armour, J. A. Espinosa, W. Money, Big data: Issues and challenges moving forward, in: 46th Hawaii International Conference on System Sciences (HICSS), IEEE, 2013, pp. 995–1004.
- [28] P. Fens, M. Funk, Personal health data: visualization modalities and their perceived values, in: Proceedings of the 22nd international confer-

ence on centre European computer graphics visualization and computer visualization, 2014, pp. 339–344.

- [29] M. Z. Sulthana, S. Habeeba, Assurance of patient control towards personal health data, International Journal of Advanced Research in Engineering and Science 2 (2014) 1660–1665.
- [30] A. J. Greenberg, A. L. Falisi, L. J. F. Rutten, W.-Y. S. Chou, V. Patel, R. P. Moser, B. W. Hesse, Access to Electronic Personal Health Records Among Patients With Multiple Chronic Conditions: A Secondary Data Analysis, Journal of Medical Internet Research 19 (2017) e188.
- [31] M. H. Aboelfotoh, P. Martin, H. S. Hassanein, A mobile-based architecture for integrating personal health record data, in: IEEE 16th International Conference on e-Health Networking, Applications and Services (Healthcom), IEEE, 2014, pp. 269–274.
- [32] A. Zanutto, Personal Health Data in Frequent Users Life: From Institutional Design to Self-tracking, in: 6th STS Italia Conference Sociotechnical Environments, 2016, p. 499.
- [33] D. Bauer, D. M. Blough, A. Mohan, Redactable signatures on data with dependencies and their application to personal health records, in: 8th ACM workshop on Privacy in the electronic society, ACM, 2009, pp. 91–100.
- [34] M. Mammadova, The Problems of Information Security of Electronic Personal Health Data, in: 7th International Conference on Information

Technology in Medicine and Education (ITME), IEEE, 2015, pp. 678–682.

- [35] K. Ishikawa, H. Ohmichi, Y. Umesato, H. Terasaki, H. Tsukuma, N. Iwata, T. Tanaka, A. Kawamura, K. Sakata, T. Sainohara, et al., The guideline of the personal health data structure to secure safety healthcare: The balance between use and protection to satisfy the patients' needs, International Journal of Medical Informatics 76 (2007) 412–418.
- [36] M. Aanestad, New Ethical Dilemmas Arising from the Growth of Personal Health Data, in: 18th Annual International Conference Dilemmas for Human Services: Organizing, Designing and Managing, 2017.
- [37] M. A. Alyami, M. Almotairi, L. Aikins, A. R. Yataco, Y.-T. Song, Managing personal health records using meta-data and cloud storage, in: IEEE/ACIS 16th International Conference on Computer and Information Science (ICIS), IEEE, 2017, pp. 265–271.
- [38] M. Li, S. Yu, K. Ren, W. Lou, Securing Personal Health Records in Cloud Computing: Patient-Centric and Fine-Grained Data Access Control in Multi-owner Settings, in: SecureComm, volume 10, Springer, 2010, pp. 89–106.
- [39] L. Chen, X. Li, Y. Yang, H. Kurniawati, Q. Z. Sheng, H.-Y. Hu, N. Huang, Personal health indexing based on medical examinations: a data mining approach, Decision Support Systems 81 (2016) 54–65.

- [40] R. Engelbrecht, L. Nicholson, Methods for Better Health: Big Data, Personal Health and More, IJBH 3 (2015) 45–51.
- [41] F. Camous, D. McCann, M. Roantree, Capturing personal health data from wearable sensors, in: International Symposium on Applications and the Internet SAINT, IEEE, 2008, pp. 153–156.
- [42] B. Moulton, Z. Chaczko, M. Karatovic, Data fusion and aggregation methods for pre-processing ambulatory monitoring and remote sensor data for upload to personal electronic health records, International Journal of Digital Content Technology and its Applications 3 (2009) 120–127.
- [43] R. Hess, C. L. Bryce, S. Paone, G. Fischer, K. M. McTigue, E. Olshansky, S. Zickmund, K. Fitzgerald, L. Siminerio, Exploring challenges and potentials of personal health records in diabetes self-management: implementation and initial assessment, Telemedicine and e-Health 13 (2007) 509–518.
- [44] M. A. Earnest, S. E. Ross, L. Wittevrongel, L. A. Moore, C.-T. Lin, Use of a patient-accessible electronic medical record in a practice for congestive heart failure: patient and physician experiences, Journal of the American Medical Informatics Association 11 (2004) 410–417.
- [45] W. Tuil, A. Ten Hoopen, D. D. Braat, P. de Vries Robbé, J. Kremer, Patient-centred care: using online personal medical records in IVF practice, Human Reproduction 21 (2006) 2955–2959.

- [46] B. A. Kitchenham, D. Budgen, O. P. Brereton, Using mapping studies as the basis for further research–a participant-observer case study, Information and Software Technology 53 (2011) 638–651.
- [47] D. Budgen, P. Brereton, Performing systematic literature reviews in software engineering, in: Proceedings of the 28th International Conference on Software Engineering, ACM, 2006, pp. 1051–1052.
- [48] A. Davis, O. Dieste, A. Hickey, N. Juristo, A. M. Moreno, Effectiveness of requirements elicitation techniques: Empirical results derived from a systematic review, in: 14th IEEE International Conference, Requirements Engineering, IEEE, 2006, pp. 179–188.
- [49] A. Roehrs, C. A. da Costa, R. da Rosa Righi, K. S. F. de Oliveira, Personal health records: a systematic literature review, Journal of Medical Internet Research 19 (2017) e13.
- [50] M.-P. Gagnon, Electronic Personal Health Records in Canada: An Exploratory Review of the Literature, in: Medicine 2.0 Conference, JMIR Publications Inc., Toronto, Canada, 2014.
- [51] S. Pushpangadan, C. Seckman, Consumer perspective on personal health records: a review of the literature, Online Journal of Nursing Informatics 19 (2015).
- [52] K. Petersen, R. Feldt, S. Mujtaba, M. Michael, Systematic Mapping Studies in Software engineering, in: In Proceedings of the 12th International Conference on Evaluation and Assessment in Software Engi-

neering, EASE'08, Blekinge Institute of Technology, Bari, Italy, 2008, pp. 71–80.

- [53] P. Brereton, B. A. Kitchenham, D. Budgen, M. Turner, M. Khalil, Lessons from applying the systematic literature review process within the software engineering domain, Journal of Systems and Software 80 (2007) 571–583.
- [54] M. Jorgensen, M. Shepperd, A Systematic Review of Software Development Cost Estimation Studies, IEEE Transactions on Software Engineering 33 (2007) 33–53.
- [55] S. Ouhbi, A. Idri, J. L. Fernandez-Aleman, T. Ambrosio, Software Quality Requirements: A Systematic Mapping Study, in: 20th Asia-Pacific Software Engineering Conference, APSEC, 2013, pp. 231–238.
- [56] S. Ouhbi, A. Idri, J. L. F. Alemán, A. Toval, Evaluating software product quality: A systematic mapping study, in: Joint Conference of the International Workshop on Software Measurement and the International Conference on Software Process and Product Measurement (IWSM-MENSURA), IEEE, 2014, pp. 141–151.
- [57] J. Street, Personal Data Protection in Health and Social Services, Journal of Medical Ethics 17 (1991) 53.
- [58] K. Ishikawa, M. Miyaji, Y. Nakamura, S. Nishi, T. Sasaki, K. Tsuji, K. Yamamoto, R. Watanabe, Troubles about Privacy Protection as Security Issues of the Personal Health Data Management in Japan and

some Solution Proposals, Studies in Health Technology and Informatics (1995) 151–154.

- [59] D. E. Detmer, E. B. Steen, Shoring up protection of personal health data, Issues in Science and Technology 12 (1996) 73–78.
- [60] J. Day, Privacy and personal health data in cyberspace, The Journal of Contemporary Dental Practice 2 (2001) 45–56.
- [61] N. Black, Secondary use of personal data for health and health services research: why identifiable data are essential, Journal of Health Services Research & Policy 8 (2003) 36–40.
- [62] T. Mallaug, K. Bratbergsengen, Long-term temporal data representation of personal health data, in: East European Conference on Advances in Databases and Information Systems, Springer, 2005, pp. 379–391.
- [63] W. Stuta, F. Wartenaa, M. van Steenb, A distributed shared data space for personal health systems, in: Ubiquity: Technologies for Better Health in Aging Societies: Proceedings of MIE2006, volume 124, IOS Press, 2006, p. 57.
- [64] B. Brown, The Number of Online Personal Health Records Is Growing, but Is the Data in These Records Adequately Protected?, Journal of Health Care Compliance 9 (2007) 35.
- [65] A. G. Matthew, K. L. Currie, J. Irvine, P. Ritvo, D. Santa Mina, L. Jamnicky, R. Nam, J. Trachtenberg, Serial personal digital assistant

data capture of health-related quality of life: a randomized controlled trial in a prostate cancer clinic, Health and Quality of Life Outcomes 5 (2007) 38.

- [66] J. H. Frost, M. P. Massagli, Social uses of personal health information within PatientsLikeMe, an online patient community: what can happen when patients have access to one another's data, Journal of Medical Internet Research 10 (2008) e15.
- [67] A. Mouttham, L. Peyton, B. Eze, A. Saddik, Event-driven data integration for personal health monitoring, Journal of Emerging Technologies in Web Intelligence 1 (2009) 110–118.
- [68] F. Andry, G. Naval, D. Nicholson, M. Lee, I. Kosoy, L. Puzankov, Data Visualization in a Personal Health Record using Rich Internet Application Graphic Components, in: HEALTHINF, 2009, pp. 111– 116.
- [69] L. Constantinescu, J. Kim, D. Feng, Integration of interactive biomedical image data visualisation to Internet-based Personal Health Records for handheld devices, in: 11th International Conference on e-Health Networking, Applications and Services, IEEE, 2009, pp. 79–83.
- [70] D. Pullman, S. K. Buehler, L. Felt, K. Gallagher, J. House, T. M. Keough, L. McDonald, A. Power, A. Ryan, Sorry, you can't have that information: data holder confusion regarding privacy requirements for personal health information and the potential chilling effect on health research, Healthcare Policy 4 (2009) 61.

- [71] E. R. Weitzman, L. Kaci, K. D. Mandl, Sharing medical data for health research: the early personal health record experience, Journal of Medical Internet Research 12 (2010) e14.
- [72] D. Capozzi, G. Lanzola, A Data Synchronization Framework for Personal Health Systems, in: International Conference on Wireless Mobile Communication and Healthcare, Springer, 2011, pp. 300–304.
- [73] S.-W. Wu, P.-H. Cheng, W.-C. Chiang, J.-K. Lin, J.-S. Lai, Categorized level management agent with forest-based data structures for accessing personal health records, in: IEEE Region 10 Conference In TENCON, IEEE, 2011, pp. 1207–1211.
- [74] I. C. Señor, J. L. F. Alemán, A. Toval, Personal health records: New means to safely handle health data?, Computer 45 (2012) 27–33.
- [75] P. Van Gorp, M. Comuzzi, Lifelong personal health data and application software via virtual machines in the cloud, IEEE Journal of Biomedical and Health Informatics 18 (2014) 36–45.
- [76] N. Huba, Y. Zhang, Designing patient-centered personal health records (PHRs): health care professionals perspective on patient-generated data, Journal of Medical Systems 36 (2012) 3893–3905.
- [77] E. R. Weitzman, S. Kelemen, L. Kaci, K. D. Mandl, Willingness to share personal health record data for care improvement and public health: a survey of experienced personal health record users, BMC Medical Informatics and Decision Making 12 (2012) 39.

- [78] M. Roantree, J. Shi, P. Cappellari, M. F. OConnor, M. Whelan, N. Moyna, Data transformation and query management in personal health sensor networks, Journal of Network and Computer Applications 35 (2012) 1191–1202.
- [79] S. Schudy, V. Utikal, The influence of (im) perfect data privacy on the acquisition of personal health data, Technical Report, 330 Wirtschaft, 2012.
- [80] J. Gladwin, Patients welcome access to online health records: Jacqui Gladwin argues that giving people the right to see their personal medical data benefits them and the healthcare professionals involved, Primary Health Care 22 (2012) 10–10.
- [81] G. Laurie, N. Sethi, Towards Principles–Based Approaches to Governance of Health–Related Research Using Personal Data, European Journal of Risk Regulation 4 (2013) 43–57.
- [82] L. Jalali, R. Jain, Building health persona from personal data streams, in: PDM@ ACM Multimedia, 2013, pp. 19–26.
- [83] K. T. Pickard, M. Swan, Big desire to share big health data: A shift in consumer attitudes toward personal health information, in: AAAI Spring Symposium Series, 2014, pp. 2168–7161.
- [84] Y. Li, L. Guo, C. Wu, C.-H. Lee, Y. Guo, Building a cloud-based platform for personal health sensor data management, in: IEEE-EMBS International Conference on Biomedical and Health Informatics (BHI), IEEE, 2014, pp. 223–226.

- [85] M. Kay, Challenges in personal health tracking: The data isn't enough, XRDS: Crossroads, The ACM Magazine for Students 21 (2014) 32–37.
- [86] K. T. Pickard, Exploring markets of data for personal health information, in: IEEE International Conference on Data Mining Workshop (ICDMW), IEEE, 2014, pp. 477–480.
- [87] S. Alabbasi, A. Ahmed, A. Rebeiro-Hargrave, K. Kaneko, A. Fukuda, Design of a personal health book for ensuring healthcare data portability, in: International Conference on Applied and Theoretical Information Systems Research (ATISR), 2014.
- [88] S. Alabbasi, A. Ahmed, K. Kaneko, A. Rebeiro-Hagrave, A. Fukuda, Data types managed database design for dynamic content: A database design for Personal Health Book system, in: IEEE Region 10 Conference TENCON, IEEE, 2014, pp. 1–5.
- [89] J. Vahidhunnisha, T. Balasubramaniam, et al., Improving Privacy in Sharing of Personal Health Data Storage on Cloud, IJITR 2 (2014) 940–944.
- [90] R. Yamamoto, Management System of Personal Data Protection in the Health Care Field, Rinsho byori. The Japanese Journal of Clinical Pathology 62 (2014) 1129–1134.
- [91] M. Almalki, K. Gray, F. M. Sanchez, The use of self-quantification systems for personal health information: big data management activities and prospects, Health Information Science and Systems 3 (2015) S1.

- [92] M. J. Bietz, C. S. Bloss, S. Calvert, J. G. Godino, J. Gregory, M. P. Claffey, J. Sheehan, K. Patrick, Opportunities and challenges in the use of personal health data for health research, Journal of the American Medical Informatics Association 23 (2015) e42–e48.
- [93] D. Blumenthal, D. McGraw, Keeping personal health information safe: the importance of good data hygiene, JAMA 313 (2015) 1424–1424.
- [94] H. Hopia, M. Punna, T. Laitinen, E. Latvala, A patient as a self-manager of their personal data on health and disease with new technology-challenges for nursing education, Nurse Education Today 35 (2015) e1-e3.
- [95] T. Sztyler, J. Völker, J. Carmona Vargas, O. Meier, H. Stuckenschmidt, Discovery of personal processes from labeled sensor data: An application of process mining to personalized health care, in: Proceedings of the International Workshop on Algorithms & Theories for the Analysis of Event Data, CEUR-WS.org, 2015, pp. 31–46.
- [96] H. J. Choi, M. J. Lee, C.-M. Choi, J. Lee, S.-Y. Shin, Y. Lyu, Y. R. Park, S. Yoo, Establishing the role of honest broker: bridging the gap between protecting personal health data and clinical research efficiency, PeerJ 3 (2015) e1506.
- [97] S. Schudy, V. Utikal, Does imperfect data privacy stop people from collecting personal health data?, Games 9 (2015) 14.
- [98] R. Ssembatya, A. V. Kayem, Secure and Efficient Mobile Personal Health Data Sharing in Resource Constrained Environments, in: IEEE

29th International Conference on Advanced Information Networking and Applications Workshops (WAINA), IEEE, 2015, pp. 411–416.

- [99] Y. Yu, X. Li, G. Hu, H. Liu, J. Mei, Y. Ni, G. Tang, G. Xie, W. Xu, Personal Health Self-Management in a Data Perspective, Studies in Health Technology and Informatics 245 (2017) 1244–1244.
- [100] K. Spencer, C. Sanders, E. A. Whitley, D. Lund, J. Kaye, W. G. Dixon, Patient perspectives on sharing anonymized personal health data using a digital system for dynamic consent and research feedback: a qualitative study, Journal of Medical Internet Research 18 (2016) e66.
- [101] J. Chen, A. Bauman, M. Allman-Farinelli, A study to determine the most popular lifestyle smartphone applications and willingness of the public to share their personal data for health research, Telemedicine and e-Health 22 (2016) 655–665.
- [102] D. Thilakanathan, R. A. Calvo, S. Chen, S. Nepal, N. Glozier, Facilitating secure sharing of personal health data in the cloud, JMIR Medical Informatics 4 (2016).
- [103] L. Koumakis, H. Kondylakis, M. Chatzimina, G. Iatraki, P. Argyropaidas, E. Kazantzaki, M. Tsiknakis, S. Kiefer, K. Marias, Designing smart analytical data services for a personal health framework, in: pHealth, 2016, pp. 123–128.
- [104] E. M. Piras, The Changing Locus of Health Data Production and Use: Patient-Generated Health Data, Observations of Daily Living,

and Personal Health Information Management, CoRR abs/1606.09589 (2016).

- [105] E. J. Beck, W. Gill, P. R. De Lay, Protecting the confidentiality and security of personal health information in low-and middle-income countries in the era of SDGs and Big Data, Global health action 9 (2016) 32089.
- [106] Y.-R. Lee, Y.-C. Chung, J.-S. Kim, H.-K. Park, Personal Health Information De-identified Performing Methods in Big Data Environments, International Journal of Software Engineering and Its Applications 10 (2016) 127–138.
- [107] A. M. Barbarin, P. Klasnja, T. C. Veinot, Good or bad, ups and downs, and getting better: Use of personal health data for temporal reflection in chronic illness, International Journal of Medical Informatics 94 (2016) 237–245.
- [108] P. Ruotsalainen, B. Blobel, Trust Model for Protection of Personal Health Data in a Global Environment, Studies in Health Technology and Informatics 245 (2017) 202–206.
- [109] A. Zanutto, Two clicks and Im in!Patients as co-actors in managing health data through a personal health record infrastructure, Health Informatics Journal (2017).
- [110] A. Troiano, Wearables and Personal Health Data: Putting a Premium on Your Privacy, Brooklyn Law Review 82 (2017) 6.

- [111] J. S. Winter, E. Davidson, Investigating Values in Personal Health Data Governance Models, in: Healthcare Informatics & Health Information Technology (SIGHealth), 2017.
- [112] P. Dabrowska-Klosińska, Tracing individuals under the EU regime on serious, cross-border health threats: An appraisal of the system of personal data protection, European Journal of Risk Regulation 8 (2017) 700–722.
- [113] S. Agboola, S. Golas, N. Fischer, M. Nikolova-Simons, J. op den Buijs, L. Schertzer, J. Kvedar, K. Jethwani, Healthcare utilization in older patients using personal emergency response systems: an analysis of electronic health records and medical alert data, BMC Health Services Research 17 (2017) 282.
- [114] N. Kirimlioglu, The right to privacy and the patient views in the context of the personal data protection in the field of health, Biomedical Research 28 (2017) 1464–1471.
- [115] D. Mendelson, Legal protections for personal health information in the age of Big Data–a proposal for regulatory framework, Ethics, Medicine and Public Health 3 (2017) 37–55.
- [116] P. Staccini, L. Fernandez-Luque, Secondary Use of Recorded or Selfexpressed Personal Data: Consumer Health Informatics and Education in the Era of Social Media and Health Apps, Yearbook of Medical Informatics 26 (2017) 172–175.

- [117] C. N. Daley, E. M. Chen, A. E. Roebuck, R. R. Ghahari, A. F. Sami, C. G. Skaggs, M. D. Carpenter, M. J. Mirro, T. R. Toscos, Providing Patients with Implantable Cardiac Device Data through a Personal Health Record: A Qualitative Study, Applied Clinical Informatics 8 (2017) 1106–1116.
- [118] A. A. B. S. Mohamed, L. F. Chen, Data Privacy Protection: A Study on Students Awareness of Personal Data Privacy Protection in an E-Health Environment, Advanced Science Letters 23 (2017) 5299–5303.
- [119] T. Huhtala, M. Pikkarainen, S. Saraniemi, Transformation of the Business Model in an Occupational Health Care Company Embedded in an Emerging Personal Data Ecosystem: A Case Study in Finland, World Academy of Science, Engineering and Technology, International Journal of Social, Behavioral, Educational, Economic, Business and Industrial Engineering 9 (2017) 3469–3477.
- [120] R. McCartney, J. Tenenberg, From Conference to Journal, Journal on Educational Resources in Computing 8 (2008).
- [121] L. V. Knight, T. A. Steinbach, Selecting an Appropriate Publication Outlet: A Comprehensive Model of Journal Selection Criteria for Researchers in a Broad Range of Academic Disciplines, International Journal of Doctoral Studies 3 (2008) 60–81.
- [122] S. K. Donovan, The importance of resubmitting rejected papers, Journal of Scholarly Publishing 38 (2007) 151–155.

- [123] M. Montesi, J. M. Owen, From conference to journal publication: How conference papers in software engineering are extended for publication in journals, Journal of the Association for Information Science and Technology 59 (2008) 816–829.
- [124] Organisation for Economic Co-operation and Development, Strengthening health information infrastructure for health care quality governance: good practices, new opportunities and data privacy protection challenges, OECD Health Policy Studies. Paris (2013).
- [125] World Health Organization, Global Diffusion of eHealth: Making Universal Health Coverage Achievable: Report of the Third Global Survey on eHealth, Geneva, Switzerland: World Health Organization (2016).
- [126] M. Foundation, Markle Foundation's Personal Health Working Group, Connecting for Health, [Accessed on May 2018]. URL: https://goo.gl/fKdAvZ.
- [127] AHIMA, Defining the personal health record, Journal of AHIMA 76 (2005) 24–25.
- [128] C. Pagliari, D. Detmer, P. Singleton, Potential of electronic personal health records, BMJ: British Medical Journal 335 (2007) 330.
- [129] PatientsLikeMe, [Accessed on May 2018]. URL: https://goo.gl/9BD2ny.
- [130] Microsoft Health Vault, [Accessed on May 2018]. URL: https://goo.gl/fZMBMp.

- [131] Google in Health, [Accessed on May 2018]. URL: https://goo.gl/paehpS.
- [132] S. Davey, M. Brennan, B. Meenan, R. McAdam, Innovative Business Models as a Tool to Manage the Emergence of Connected Health Technologies, in: Irish Academy of Management Conference, Irish Academy of Management, Blackwell Publishing, 2010.
- [133] I. G. Chouvarda, D. G. Goulis, I. Lambrinoudaki, N. Maglaveras, Connected health and integrated care: Toward new models for chronic disease management, Maturitas 82 (2015) 22–27.
- [134] J. Kvedar, M. J. Coye, W. Everett, Connected health: a review of technologies and strategies to improve patient care with telemedicine and telehealth, Health Affairs 33 (2014) 194–199.
- [135] D. E. Bloom, D. Canning, G. Fink, Implications of population ageing for economic growth, Oxford Review of Economic Policy 26 (2010) 583-612.
- [136] D. D. Vergados, Service personalization for assistive living in a mobile ambient healthcare-networked environment, Personal and Ubiquitous Computing 14 (2010) 575–590.
- [137] IDC, Smartphone OS Market Share, 2017 Q1, [Accessed on May 2018].URL: https://goo.gl/Np6u2i.
- [138] Statista, Market share worldwide 2009-2015-Statistic, [Accessed on May 2018]. URL: https://goo.gl/5nPBmL.

- [139] B. Wire, Finding success in the new IoT ecosystem: market to reach 3.04 trillion and 30 billion connected Things in 2020, [Accessed on May 2018]. URL: https://goo.gl/aAZGqb.
- [140] D. Evans, The Internet of Things: how the next evolution of the Internet is changing everything, Technical Report, Cisco Internet Business Solutions Group, 2011.
- [141] Agfa HealthCare Strengthens IT Strategy by Extending Offering to Integrated Care, [Accessed on May 2018]. URL: https://goo.gl/KnM2DM.
- [142] K. Petersen, R. Feldt, S. Mujtaba, M. Mattsson, Systematic Mapping Studies in Software Engineering, in: EASE, volume 8, 2008, pp. 68–77.
- [143] R. Harte, L. Glynn, A. Rodríguez-Molinero, P. M. Baker, T. Scharf, L. R. Quinlan, G. ÓLaighin, A human-centered design methodology to enhance the usability, human factors, and user experience of connected health systems: a three-phase methodology, JMIR human factors 4 (2017) e8.
- [144] Food and Drug Administration (FDA), Draft Guidance for Industry and Food and Drug Administration Staff. Applying Human Factors and Usability Engineering to Optimize Medical Device Design, Technical Report, Food and Drug Administration, Rockville, Md, USA, 2011.
- [145] D. Armijo, C. McDonnell, K. Werner, Electronic health record usability, Technical Report 10, Agency for Healthcare Research and Quality (AHRQ), 2009.

- [146] T. Heart, O. Ben-Assuli, I. Shabtai, A review of PHR, EMR and EHR integration: A more personalized healthcare and public health policy, Health Policy and Technology 6 (2017) 20–25.
- [147] R. Steele, A. Lo, Future personal health records as a foundation for computational health, in: International Conference on Computational Science and Its Applications, Springer, 2009, pp. 719–733.
- [148] V. Venkatesh, F. D. Davis, A theoretical extension of the technology acceptance model: Four longitudinal field studies, Management science 46 (2000) 186–204.
- [149] A. Idri, S. Ouhbi, J. L. Fernández-Aléman, A. Toval, A survey of requirements engineering education, in: IEEE Global Engineering Education Conference (EDUCON), IEEE, 2012, pp. 1–5.
- [150] J. Liu, X. Huang, J. K. Liu, Secure sharing of personal health records in cloud computing: ciphertext-policy attribute-based signcryption, Future Generation Computer Systems 52 (2015) 67–76.
- [151] J. P. Albrecht, How the GDPR Will Change the World, European Data Protection Law Review 2 (2016) 287.
- [152] I. N. Shu, H. Jahankhani, The Impact of the new European General Data Protection Regulation (GDPR) on the Information Governance Toolkit in Health and Social care with special reference to Primary care in England, in: Cybersecurity and Cyberforensics Conference (CCC), IEEE, 2017, pp. 31–37.

- [153] B. M. Knoppers, A. M. Thorogood, Ethics and big data in health, Current Opinion in Systems Biology 4 (2017) 53–57.
- [154] L. L. DuBenske, D. H. Gustafson, B. R. Shaw, J. F. Cleary, Webbased cancer communication and decision making systems: connecting patients, caregivers, and clinicians for improved health outcomes, Medical Decision Making 30 (2010) 732–744.
- [155] S. C. Kim, D. V. Shah, K. Namkoong, F. M. McTavish, D. H. Gustafson, Predictors of online health information seeking among women with breast cancer: the role of social support perception and emotional well-being, Journal of Computer-Mediated Communication 18 (2013) 212–232.
- [156] D. B. Lafky, T. A. Horan, Personal health records: Consumer attitudes toward privacy and security of their personal health information, Health Informatics Journal 17 (2011) 63–71.
- [157] K. Petersen, S. Vakkalanka, L. Kuzniarz, Guidelines for conducting systematic mapping studies in software engineering: An update, Information and Software Technology 64 (2015) 1–18.
- [158] F. Elberzhager, J. Münch, V. T. N. Nha, A systematic mapping study on the combination of static and dynamic quality assurance techniques, Information and Software Technology 54 (2012) 1–15.
- [159] V. Garousi, A. Mesbah, A. Betin-Can, S. Mirshokraie, A systematic mapping study of web application testing, Information and Software Technology 55 (2013) 1374–1396.

- [160] W. M. Bramer, D. Giustini, G. B. de Jonge, L. Holland, T. Bekhuis, De-duplication of database search results for systematic reviews in End-Note, Journal of the Medical Library Association JMLA 104 (2016) 240.
- [161] A. Ampatzoglou, S. Charalampidou, I. Stamelos, Research state of the art on gof design patterns: A mapping study, Journal of Systems and Software 86 (2013) 1945–1964.
- [162] A. M. Fernández-Sáez, M. Genero, M. R. Chaudron, Empirical studies concerning the maintenance of uml diagrams and their use in the maintenance of code: A systematic mapping study, Information and Software Technology 55 (2013) 1119–1142.
- [163] S. Ouhbi, A. Idri, J. L. Fernández-Alemán, A. Toval, Requirements engineering education: a systematic mapping study, Requirements Engineering 20 (2015) 119–138.
- [164] S. Ouhbi, Software architecture evaluation: A systematic mapping study, in: Proceedings of the 13th International Conference on Evaluation of Novel Approaches to Software Engineering, ENASE 2018, Funchal, Madeira, Portugal, March 23-24, 2018., 2018, pp. 447–454.
- [165] P. A. Higgins, A. J. Straub, Understanding the error of our ways: Mapping the concepts of validity and reliability, Nursing Outlook 54 (2006) 23–29.
- [166] D. M. Rousseau, J. Manning, D. Denyer, 11 evidence in management and organizational science: Assembling the field's full weight of scien-

tific knowledge through syntheses, Academy of Management Annals 2(2008)475–515.